Author’s response to reviews

Title: SPECIALIST LINK AND PRIMARY CARE NETWORK CLINICAL PATHWAYS- A NEW APPROACH OF PATIENT REFERRAL: A CROSS-SECTIONAL SURVEY OF AWARENESS, UTILIZATION AND USABILITY AMONG FAMILY PHYSICIANS IN CALGARY

Authors:

Mubashir Arain (mubashiraslam.arain@ahs.ca)
Mahnoush Rostami (mahnoush.rastami@ahs.ca)
Mariama Zaami (mariama.zaami@ahs.ca)
Valerie Kiss (valerie.kiss@ahs.ca)
Richard Ward (richard.ward@ahs.ca)

Version: 3 Date: 29 Apr 2020

Author’s response to reviews:

Dear Editor,
Thank you for giving us the opportunity to revise the manuscript. We have addressed all comments in the revised manuscript. Our point-to-point responses are provided below.
Thank you,
Mubashir Arain

Comments
1. We note that the order of authors has changed since the original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

Response from authors:
There was a confusion about authors’ names when the manuscript revision was submit which was corrected. Now the list and sequence of author is exactly the same as it was originally submitted. Thank you.

2. In your “Ethics approval and consent to participate” section, please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If only verbal consent was obtained, please state the reason and whether the ethics committee approved this procedure.

Response from authors:
We have provided complete ethics information in the main manuscript.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response from authors:
We have uploaded the final clean version of manuscript and all figures as separate files. Thanks